Gustavo de Sousa Arantes Ferreira  
Instituto de Cardiologia do Distrito Federal  
Setor HFA - Hospital das Forças Armadas - Setor Sudoeste, S/Nº  
CEP: 70673-900804  
Brasilia-DF – Brazil  
gustferr@ufmg.br

To the reviewers of the World Journal of Hepatology

August 7, 2022

Dear sirs:

Thank you all for your valuable considerations and corrections regarding our manuscript.

Regarding the comments by reviewer number 05685221, we thank you for your feedback. We have modified our discussion section to highlight the novel aspects of this case report (mainly the use of percutaneous biopsy to confirm the diagnosis of hepatic infarction).

Regarding the comments by reviewer number 02446627, we thank you for your feedback. We have modified our history of present illness and discussion sections to stress that the patient was indeed hypotensive when he first presented to the emergency department, which has probably contributed to the development of the liver infarcts as pointed out by the reviewer.

Thank you for your consideration!

Sincerely,

Gustavo de Sousa Arantes Ferreira, Department of Liver Transplantation, Instituto de Cardiologia do Distrito Federal